Author's response to reviews

Title: Unplanned readmission rates, length of hospital stay, mortality, and medical costs of ten common medical conditions: a retrospective analysis of Hong Kong hospital data

Authors:

Eliza LY Wong (lywong@cuhk.edu.hk)
Annie WL Cheung (anniewlcheung@cuhk.edu.hk)
Michael CM Leung (cmleung123@hotmail.com)
Carrie HK Yam (carrieyam@cuhk.edu.hk)
Frank WK Chan (cwkfrank@cuhk.edu.hk)
Fiona YY Wong (fionawong@cuhk.edu.hk)
Eng Kiong Yeoh (yeoh_ek@cuhk.edu.hk)

Version: 4 Date: 10 June 2011

Author's response to reviews: see over
10 June 2011

Editor-in-Chief

BMC Health Service & Research

Dear Sir/Madam,

Re: Manuscript MS: 1987061455482610

"Unplanned readmission rates, length of hospital stay, mortality, and medical costs of ten common medical conditions: a retrospective analysis of Hong Kong hospital data"

Thank you very much again for accepting our manuscript. We have revised the manuscript in red and response it point-to-point in the following page.

Yours sincerely,

Eliza Wong
Assistant Professor
School of Public Health & Primary Care
The Chinese University of Hong Kong
Major revisions

Comment 1: Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)
Response: I have updated the authorship at page 1.

Comment 2: Authors' contributions: Please indicate that all authors read and approved the final manuscript.
Response: The sentence is added at page 17.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Comment 3: Corresponding author: Please link to author list with superscript or remove symbol
Response: The corresponding author is indicated with asterisk at page 1.

Comment 4: References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.
Response: The reference 19, 24 & 52 are revised.

Comment 5: References: please provide the full citation details for reference 6. You can view the BMC reference guide at the following link
http://www.biomedcentral.com/bmcgenet/ifora/#references
Response: The detail for reference 6 is added.

Comment 6: Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Response: The typo is checked.